Author’s response to reviews

Title: Paediatric Behcet's disease presenting with recurrent papillitis and episcleritis: a case report

Authors:

Fulvio Parentin (parentin@burlo.trieste.it)
Loredana Lepore (leporel@burlo.trieste.it)
Ingrid Rabach (ingridrabach@gmail.com)
Stefano Pensiero (pensiero@burlo.trieste.it)

Version: 3 Date: 26 July 2010

Author’s response to reviews:

Response to referee:
Referee 1:
1) HLA status was mentioned
2) Conclusions were separated from discussion

Referee 2:
1) Typographical and grammatical errors were corrected
2) “Yr” was expanded to “year”
3) Conclusions were separated from discussion
4) Status of HLA B51 was mentioned
5) The role of infliximab in Behcet's disease was discussed

Response to editorial requests:
1) I separated the conclusion section into a discussion and a conclusion.
2) Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.
3) I have highlighted with coloured text all changes made.
4) The manuscript conforms to the journal style

Dr. Fulvio Parentin
Institute For Maternal and Child Health Burlo Garofolo, Ophthalmology
Via Dell'Istria 65/1 Trieste, I-34137